Author's response to reviews

Title: Bilateral dilatation of the urinary tract due to iliopsoas pyomyositis: a case report

Authors:

Aristotelis Tsiakalos (atsiakalos@gmail.com)
Sarah P Georgiadou (sgeorg@med.uth.gr)
Ioannis Anastasiou (ekati2@otenet.gr)
Nikolaos V Sipsas (nsipsas@med.uoa.gr)
Athanasios Tzioufas (agtzi@med.uoa.gr)

Version: 2 Date: 13 December 2010

Author's response to reviews:

December 13, 2010

To

Journal of Medical Case Reports
Editorial Office

Dear Editor

We would be much obliged if you could consider the manuscript entitled" Bilateral dilatation of the urinary tract due to Iliopsoas pyomyositis: a case report” for publication in your Journal.

All authors feel that the particular findings of this manuscript are valid and will add further to currently existing knowledge. Our case report is characterized by unexpected event in the course of treating a patient with iliopsoas pyomyositis. To our knowledge, this is first report of this kind in the literature and it is an original case report of interest to different clinical specialities of medicine such as internal medicine, infectious diseases and urology.

All authors have made a significant contribution to the findings and methods in the paper and have read and approved the final draft.

The manuscript has not already been published and has not been submitted simultaneously to any other journal. There was not any financial support or other benefits from commercial sources for the work reported on in the manuscript. There was not any conflict of interest. The corresponding author takes on the above responsibilities with his signature.

Finally, in order to conform to the journal style, we followed the instructions for authors and confirmed the data about patient’s ethnicity and removed all the identifying information from figure 1.

The corresponding author is
Sarah P Georgiadou, MD
Address: 75 M Asias St, 11527, Athens, Greece
Telephone number: ++30 210 7462643
Fax number : ++30 210 7462664
E-mail address : sgeorg@med.uth.gr
Yours sincerely,
SP Georgiadou MD